Author's response to reviews

Title: Use of "biokitHSV-2 Rapid Assay" to Improve the Positive Predictive Value of Focus HerpeSelect HSV-2 ELISA

Authors:

Rhoda Ashley Morrow Ph.D. (rhoda.morrow@seattlechildrens.org)
David Friedrich BS (david.friedrich@seattlechildrens.org)
Amalia Meier Ph.D. (ameier@u.washington.edu)
Lawrence Corey MD (lcorey@u.washington.edu)

Version: 3 Date: 4 October 2005

Author's response to reviews:

We have check the Title and made sure it is the same and the Author List is the same and in the same order.

We have removed all full stops/hard returns in the Section headings and sub-headings.

We have removed the Footer from the manuscript.

We have put the Figures in the Manuscript with the legend and eliminated the title Figures from each.